Author's response to reviews

Title: Dermoscopy as a technique for the early identification of foot melanoma

Authors:

Ivan R Bristow (ib@soton.ac.uk)
Jonathan Bowling (jonathan.Bowling@orh.nhs.uk)

Version: 5 Date: 16 March 2009

Author's response to reviews:

Editorial Revision Points made (as requested):

Reference 52 is not referred to in the text but appears as an annotation on figure 4 hence we have left it in the reference list.

Author contributions have now been added in the text.

The title on table 1 has been placed above it.

Page 16 (blank) has now been removed